Author’s response to reviews

Title: Nurses' and care workers' experiences of spiritual needs in patients with dementia disease in nursing homes: a qualitative study.

Authors:

Liv Ødbehr (liv.odbehr@hihm.no)
Kari Kvigne (kari.kvigne@hihm.no)
Solveig Hauge (solveig.hauge@hit.no)
Lars Danbolt (lars.danbolt@sykehuset-innlandet.no)

Version: 4 Date: 30 July 2013

Author's response to reviews: see over
Dear Editor in chief, Catia Cornacchia  
Catia.Cornacchia@biomedcentral.com

Enclosed the manuscript with the title: “Nurses’ and care workers’ experiences of spiritual needs in patients with dementia disease in nursing homes: a qualitative study”. Please consider the manuscript for publication in BMC Nursing.

The authors guarantee that neither all nor any part of this manuscript has been published elsewhere in its present form in another publication or under a different title by either this or another author. It is also acknowledged that this manuscript is not currently under review by any other publication. The authors also declare that they have no competing interests.

Yours sincerely
Hedmark University College

On behalf of the authors’
Liv Ødbehr

RN. Lecturer. PhD student
Department of Nursing and Mental Health, Faculty of Public Health, Hedmark University College, P.O Box 400. 2418 Elverum, Norway.
Mobile Phone number: +47 959 079 84
E-mail: liv.odbehr@hihm.no
Changes are marked with red:

1. **Cover letter:**
   In the title of the study the word “disease” is added.

2. **Title page:**
   i) The name of the main author is shortened from Liv Skomakerstuen Ødbehr to:
      Liv Ødbehr.
   ii) Competing interests is added to the cover letter.
   iii) Funding statements are removed from title page to the end of main manuscript.
   iv) Title page is included in the main manuscript file, and removed from the “additional files”.

3. **Manuscript detailed tab:**
   Clinical trials only:

   Question: “Is this a clinical-trial-related article?”

   Yes

   If yes, please select the appropriate description:

   Results- for other types of trials

   Trial registration number:

   26783

   Name of register:

   Norwegian Social Science Data Service
4. **Main manuscript:**

i) Ethical consideration was taken into account relying on WMA Declaration of Helsinki (Ethical Principles for Medical Research Involving Human Subjects, http://www.wma.net/). It was obtained licenses of leaders from four municipalities and nursing institutions in Norway, and the participation was voluntarily. To ensure the participants' well-being, integrity and dignity throughout the interview process, the importance of safeguarding the confidentiality of each participant was emphasized. We obtained permission to conduct the study from Norwegian Social Science Data Services (NSD) in April 2011, with reference number 26783.

ii) The acronym of Liv (Skomakerstuen) Ødbehr, LSØ, is changed to LØ in the main manuscript under “authors’ contribution”.

5. **Acknowledgements:**

Acknowledgements are moved from the title page to the main manuscript.